Author's response to reviews

Title: TOL19-001 reduces inflammation and MMP expression in monolayer cultures of tendon cells

Authors:

Catherine BAUGE (catherine.bauge@unicaen.fr)
Sylvain LECLERCQ (sylvainleclercq@wanadoo.fr)
Thierry CONROZIER (tconrozier@chbm.fr)
Karim BOUMEDIENE (karim.boumediene@unicaen.fr)

Version: 4  Date: 31 May 2015

Author's response to reviews:

Concerning the conflict of interest, as clearly indicated in the manuscript, the funding support for this study was Labhra laboratory. However, Labhra had no role in the study design, experimentations, data collection and analysis, decision to publish, or preparation of the manuscript. Consequently, their participation as a funder did not interfere with this study, and did not influence the results or redaction of the manuscript.